Author's response to reviews

Title: 'Let's Get Moving' : Promoting Physical Activity in Primary Care

Authors:

Fiona C Bull Prof (Fiona.Bull@uwa.edu.au)
Karen Milton Miss (K.Milton@lboro.ac.uk)

Version: 2 Date: 28 May 2010

Author's response to reviews: see over
28th May 2010

Dear Dr. Norton

Thank you for the opportunity to resubmit our manuscript reporting the process evaluation of a primary care based physical activity intervention in the UK. We have considered the reviewers comments very carefully and adopted the majority of their suggestions and edited the paper accordingly.

Please find attached the revised paper. We also provide a point by point response to the three reviewers’ comments and cross reference each of our actions.

We believe this revised manuscript is much stronger and more clear. This area of research is particularly important given the recent Governments commitment to increasing physical activity in the UK and the inclusion of approaches in primary care as part of the Be Active Be Healthy policy released last year.

We hope you view this submission as timely and we look forward to your response.

Yours Sincerely,

Professor Fiona Bull
Director BHF National Centre for Physical Activity
School of Sport, Exercise and Health Sciences
Loughborough University